Author's response to reviews

Title: Effectiveness of the introduction of a Chronic Care Model-based program for type 2 diabetes in Belgium

Authors:

Patricia Sunaert (patricia.sunaert@ugent.be)
Hilde Bastiaens (hilde.bastiaens@ua.ac.be)
Frank Nobels (frank.nobels@olvyz-aalst.be)
Luc Feyen (luc.feyen@ugent.be)
Geert Verbeke (geert.verbeke@med.kuleuven.be)
Etienne Vermeire (etienne.vermeire@ua.ac.be)
Jan De Maeseneer (jan.demaeseneer@ugent.be)
Sara Willems (sara.willems@ugent.be)
An De Sutter (an.desutter@ugent.be)

Version: 5 Date: 8 July 2010

Author's response to reviews: see over
Author’s response to the editor

Title: Effectiveness of the introduction of a Chronic Care Model-based program for type 2 diabetes in Belgium.

Authors:

Patricia Sunaert (patricia.sunaert@ugent.be)
Hilde Bastiaens (hilde.bastiaens@ua.ac.be)
Frank Nobels (frank.nobels@olvz-aalst.be)
Luc Feyen (luc.feyen@ugent.be)
Geert Verbeke (geert.verbeke@med.kuleuven.be)
Etienne Vermeire (etienne.vermeire@ua.ac.be)
Jan De Maeseneer (jan.demaeseneer@ugent.be)
Sara Willems (sara.willems@ugent.be)
An De Sutter (an.desutter@ugent.be)

Version: 4 Date: 8 July 2010

Author’s response:

Dear Editor,

I have addressed the formatting changes as requested.

Please do not hesitate to contact me if additional information is required.

With kind regards,

Patricia Sunaert
Formatting changes to be made:

Major revisions (we require the author to make these changes)

- References 14, 15 and 19: please provide a functioning link to website
  Response: Excuse me for this inconvenience. I have adapted the links.

- Reference 15: Please provide a website title for this website.
  Response: I have added a website title as requested.

- Tables 1 and 2: We notice you have included bulleted lists as tables, please reformat so it is incorporated within the main text. Please remember to update any figure/ additional file capitations affected.
  Response: I prefer to present this information in a table given the fact that a presentation in a box is not possible. I have adapted the layout; the information is now presented in a table (tables 1 and 2).

- References: References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example 'text [1].' Reference 37 is currently cited out of numerical order in the manuscript.
  Response: The numerical order is adapted.